Author's response to reviews

Title: Different Pharmacological activity (Antioxidant, Phytochemical Screening, Thrombolytic, Antibacterial & Antifungal Potential and Cytotoxic) of Clausena heptaphylla Ethanolic stem bark extract.

Authors:

Md. Fakruddin (fakruddinmurad@gmail.com)
Reaz Md. Mazumdar (reazbio@gmail.com)
Hafsa Afroz (hafsa.afroz@gmail.com)
Murshed H Sarker (bacilluskhan@yahoo.com)

Version: 2 Date: 18 May 2012

Author's response to reviews: see over
15/05/2012
To
Editor
BMC Complementary and Alternative Medicine

Subject: SUBMISSION OF A MANUSCRIPT FOR EVALUATION

Dear Editor

I am enclosing herewith a manuscript entitled "Different Pharmacological activity (Antioxidant, Phytochemical Screening, Thrombolytic, Antibacterial & Antifungal Potential and Cytotoxic) of Clausena heptaphylla Ethanolic stem bark extract" for publication in "BMC Complementary and Alternative Medicine" for possible evaluation.

With the submission of this manuscript I would like to undertake that the above mentioned manuscript has not been published elsewhere, accepted for publication elsewhere or under editorial review for publication elsewhere.

Sincerely yours

Md. Fakruddin